Author's response to reviews

Title: Pre-hospital care time intervals among victims of road traffic injuries in Iran. A cross-sectional study

Authors:
Maryam Bigdeli (maryam.bigdeli.333@student.ki.se)
Davoud Khorasani-Zavareh (davoud.khorasani@yahoo.co.uk)
Reza Mohammadi (reza_mohammadi@hotmail.com)

Version: 5 Date: 6 July 2010

Author's response to reviews: see over
Dear Editor-in-Chief,

Please find attached our article entitled “Pre-hospital care time intervals among victims of road traffic injuries in Iran. A cross-sectional study”.

Thank you so much for accepting our manuscript for publication in BMC Public Health. We checked the manuscript as well as its format once more and we fixed everything. Now it is ready for publication.

Yours sincerely,

Davoud Khorasani-Zavareh

On behalf of all co-authors